Author's response to reviews

Title: A Systematic Review of How Homeopathy is represented in Conventional and CAM Peer Reviewed Journals

Authors:

Timothy Caulfield (tcaulfld@law.ualberta.ca)
Suzanne Debow (sdebow@ualberta.ca)

Version: 4 Date: 9 June 2005

Author's response to reviews:

Most of the reviewer's comments deal with the term "bias." The reviewer believes that a bias must intentional. However, as we note in the paper, a bias can be the result of systemic pressures and need not be intentional or the product of an explicit policy.

As requested, we removed the last sentence of the abstract. In addition, we made all revisions requested by the editor.

Yours truly,

Timothy Caulfield